Author’s response to reviews

Title: CD14+ macrophages that accumulate in the colon of African AIDS patients express pro-inflammatory cytokines and are responsive to lipopolysaccharide

Authors:

Edana Cassol (edana.cassol@carleton.ca)
Theresa Rossouw (theresa.rossouw@up.ac.za)
Susan Malfeld (susan.malfeld@up.ca.za)
Phetole Mahasha (thakgalo03@gmail.com)
Tomas Slavik (slavikt@ampath.co.za)
Chris Seebregts (chris@jembi.org)
Robert Bond (robertbond@mweb.co.za)
Johannie du Plessis (johannie.duplessis@kuleuven.be)
Carl Janssen (carljanssen@uz.leuven.be)
Tania Roskams (Tania.Roskams@med.kuleuven.be)
Frederik Nevens (frederik.nevens@uz.kuleuven.ac.be)
Massimo Alfano (alfano.massimo@hsr.it)
Guido Poli (poli.guido@hsr.it)
Schalk van der Merwe (schalk.vandermerwe@uzleuven.be)

Version: 2 Date: 27 March 2015

Author’s response to reviews:

As requested by the Editors, we added the line and page numbers. We also reformatted the Authors’ contribution section as requested. This section can be found immediately following the Competing Interests Section.